Author’s response to reviews


Authors:
Yassine Bouatou (yrbouatou@gmail.com)
Olivia Seyde (olivia.seyde@hcuge.ch)
Solange Moll (solange.moll@hcuge.ch)
Pierre-Yves Martin (pierre-yves.martin@hcuge.ch)
Jean Villard (jean.villard@hcuge.ch)
Sylvie Ferrari-Lacraz (sylvie.ferrari@hcuge.ch)
Karine Hadaya (karine.hadaya@hcuge.ch)

Version: 2 Date: 20 Mar 2018

Author’s response to reviews:

Dear Editor,

thank you for giving us the opportunity to revise our manuscript. The manuscript has been sent to AJE for English editing.

As for the cumulative incidence of DSA, it can be found in the methods section: 15%

We hope that following our answers, you'll find our manuscript suitable for publication.

On behalf of the authors,

Yassine Bouatou